

Identification and characterization of unknown disturbances in a structured population using high-throughput phenotyping data and measurement of robustness: application to growing pigs

Vincent Le, Tom Rohmer, Ingrid David

▶ To cite this version:

Vincent Le, Tom Rohmer, Ingrid David. Identification and characterization of unknown disturbances in a structured population using high-throughput phenotyping data and measurement of robustness: application to growing pigs. Journal of Animal Science, 2024, 10.1093/jas/skae059. hal-04491582

HAL Id: hal-04491582 https://hal.inrae.fr/hal-04491582

Submitted on 6 Mar 2024

HAL is a multi-disciplinary open access archive for the deposit and dissemination of scientific research documents, whether they are published or not. The documents may come from teaching and research institutions in France or abroad, or from public or private research centers. L'archive ouverte pluridisciplinaire **HAL**, est destinée au dépôt et à la diffusion de documents scientifiques de niveau recherche, publiés ou non, émanant des établissements d'enseignement et de recherche français ou étrangers, des laboratoires publics ou privés.



Distributed under a Creative Commons Attribution 4.0 International License

Identification and characterization of unknown disturbances in a structured population using high-throughput phenotyping data and measurement of robustness: application to growing pigs

Vincent Le^{*,†}, Tom Rohmer^{*}, Ingrid David^{*}

^{*} GenPhySE, Université de Toulouse, INRAE, ENVT, F-31326 Castanet Tolosan, France [†] Alliance R&D, 35650 Le Rheu, France

> Corresponding author: Ingrid David Email: Ingrid.david@inrae.fr

© The Author(s) 2024. Published by Oxford University Press on behalf of the American Society of Animal Science.

çcet

This is an Open Access article distributed under the terms of the Creative Commons Attribution License (https://creativecommons.org/licenses/by/4.0/), which permits unrestricted reuse, distribution, and reproduction in any medium, provided the original work is properly cited.

Lay Summary: Improving the response of animals to environmental disturbances in terms of robustness is a key element to face the new breeding constraints related to climate change and the agroecological transition. Characterizing the disturbances that an animal experiences is a necessary first step to correctly evaluate its robustness. We propose a new method to do so based on the analysis of high-throughput phenotyping data. Using simulated data, we demonstrate that this method is effective for detecting and characterizing unknown disturbances and is thus helpful to correctly evaluate an animal's robustness. Applied to real growing pig data, it allowed us to obtain new measurements of robustness and to estimate their heritability in order to consider including these new traits for selection.

Teaser Text: The UpDown method is an efficient new tool to detect and characterize unknown disturbances using high-throughput phenotyping data. It also makes it possible to obtain new measurements of animal robustness.

Abstract

Improving the robustness of animals has become a priority in breeding due to climate change, new societal demands and the agroecological transition. Components of animal robustness can be extracted from the analysis of the adaptive response of an animal to disturbance using longitudinal data. Nonetheless, this response is a function of animal robustness as well as of disturbance characteristics (intensity and duration). To correctly assess an animal's robustness potential, it is therefore useful to know the characteristics of the disturbances it faces. The UpDown method, which detects and characterizes unknown disturbances at different levels of organization of the population (e.g., individual, pen, batch disturbances), has been proposed for this purpose. Furthermore, using the outputs of the method, it is possible to extract proxies of the robustness of animals. In this context, the objective of the study was to evaluate the performances of the UpDown method to detect and characterize disturbances and to quantify the robustness of animals in a genetic framework using different sets of simulation, and to apply this method to real pig longitudinal data recorded during the fattening period (body weight, cumulative feed intake and feeding rate). Based on the simulations, the specificity of the UpDown method was high (> 0.95). Its sensitivity increased with the level of organization exposed (from 0.23 to 0.32 for individual disturbances, from 0.45 to 0.59 for pen disturbances, and from 0.77 to 0.88 for batch disturbances). The UpDown method also showed a good ability to characterize detected disturbances. The average time interval between the estimated and true start date or duration of the disturbance was lower than three days. The correlation between the true and estimated intensity of the disturbance increased with the hierarchical level of organization (on average, 0.41, 0.78 and 0.83 for individual, pen and batch disturbance, respectively). The accuracy of the estimated breeding values of the proxies for robustness extracted from the analysis of individual trajectories over time were moderate (lower than 0.33). Applied to real data, the UpDown method detected different disturbances depending on the phenotype analyzed. The heritability of the proxies of robustness were low to moderate (ranging from 0.11 to 0.20).

Keywords

Disturbances detection, Longitudinal phenotype, Pig, Robustness

List of Abbreviations:

ABT: area between trajectories ACF: automatic concentrate feeder ADFI: average daily feed intake ADG: average daily gain AFR: average feed rate BW: body weight CFI: cumulative feed intake FR: feed rate LW: large white Smin: slope minimum Sresil: slope for resilience th: theoretical disturbed elements THI: temperature humidity index

Reel

nusci

Introduction

Due to climate change that will inevitably lead to more extreme events (Pasqui and Di Giuseppe, 2019), new societal demands that imply changes in breeding systems (Alonso et al., 2020), and the agroecological transition associated with greater feed variability (Phocas et al., 2014), livestock will undergo more and more disturbances of various origins. In this context, improving the robustness of animals, defined as "the ability to combine a high production potential with resilience to stressors, allowing for unproblematic expression of a high production potential in a wide variety of environmental conditions" (Knap, 2005), has become a priority in breeding. Two options for breeding for animal robustness are considered in the literature (Rauw and Gomez-Rava, 2015). The first one consists in directly including measurable robustness traits in the breeding objective. However, robustness is a complex trait that is difficult to phenotype (Friggens et al., 2017). Today, traits included in breeding goals that may be associated with robustness are mainly related to the health status of animals during a specific period or to the longevity of reproductive animals (Knap, 2005; Berghof et al., 2019; Knap and Doeschl-Wilson, 2020; Lenoir et al., 2022). The second option consists in estimating the environmental sensitivity of animals by modeling and integrating the corresponding genetic value into the selection objective. The most commonly used model for this is the reaction norm model, which necessitates the environment to be described in the form of a gradient (Falconer, 1990; Kolmodin et al., 2002; Calus et al., 2004). More sophisticated models have recently been proposed, using high-throughput phenotyping data to study the dynamic nature of animals' adaptive responses to disturbance (Codrea et al., 2011; Sadoul et al., 2015; Revilla et al., 2019; Nguyen-Ba et al., 2020; Poppe et al., 2020). Some of these methods require the disturbance to be identified (e.g., weaning stress), while others do not but assume that animals reared together undergo the same disturbance, which makes their response to the disturbance comparable. However, disturbance can occur at different levels of a farm's organization and does not necessarily affect all contemporary animals. For example, based on the structure of a pig farm where animals are raised in small intra-pen groups and intra-batch groups of pens, an environmental disturbance such as a heat wave will affect all the animals in the batch, whereas a problem with the electronic feeder will only affect the individuals in the pen concerned, and a metabolic disease, for example, will only affect the animal concerned. To our knowledge, very few studies have simultaneously used information from a group of contemporary animals to detect and characterize unknown disturbances, which should nevertheless enable us to gain in detection power. Garcia-Baccino et al. (2021) proposed to detect disturbances by analyzing the daily phenotypic variance of a group of animals. Its distribution is considered as a mixture of two normal distributions (disturbed/non-disturbed days). The authors proposed using the probability of belonging to the disturbed distribution for each day as a measure of the environmental challenge that is then included as a covariate in a reaction norm animal model to estimate its environmental sensitivity. Another method to detect disturbances based on the analysis of longitudinal data at the group level was recently proposed, applicable using the "UpDown" package (David et al., 2023) in R software (R Core Team, 2021). The method analyzes the dynamics of longitudinal phenotypes at different group levels, once again using a mixture of two normal distributions to classify elements at each level of organization as disturbed or not and to characterize the detected disturbances. The aim of our study was to evaluate the performance of the UpDown method in depth via simulation in order to detect and characterize disturbances in a population structured into different group levels, and to quantify the robustness of animals in a genetic framework. This method was applied to real body weight, cumulative feed intake and the feeding rate longitudinal records of 5,872 pigs in order to perform a genetic analysis of their robustness.

Materials and Methods

Data were collected in accordance with the applicable national regulations on livestock welfare in France.

The UpDown method

Let us consider a population organized into different grouping levels of contemporary animals. For example, in the pig farming system, the population is organized into three group levels: individual, pen and batch. Each batch is made up of a group of several pens and each pen is made up of a group of several animals. In other words, the individuals are included in pens that are included in batches. The UpDown method (David et al., 2023) involves studying longitudinal observations at different group scales to facilitate the detection and characterization of disturbances on the basis of changes in the dynamics of observations over time. Indeed, Sauvant and Martin (2010) consider that the response of animals facing a disturbance can be broken down into two parts: change in state when exposed to the disturbance (part 1), and once the disturbance is over, the animal's return to its initial state (if it is elastic, part 2). A schema describing changes in state for an elastic system is provided in Figure 1. This dynamic response of an animal to a disturbance can be described in terms of resistance and resilience, which are defined as the capacity of an animal to minimize impacts of perturbing factors and to quickly return to the pre-perturbed condition (Sadoul et al., 2015; Nguyen-Ba et al., 2020). When the disturbance appears, the importance of change in state will depend on the intensity of the disturbance, the resistance of the animal, as well as on its resilience as soon as the animal has quit its initial state to limit the effect of the perturbing factor. When the disturbance is over, only the resilience capacity is involved until the initial state is restored. The UpDown method consists of two steps: (i) the Up-step identifies the elements facing disturbance at each group level, starting at the individual level and ending at the highest organizational level (i.e., the batch in the pig system); and (ii) the Down-step validates elements detected in the Up-step, from the highest to the individual level of organization, and characterizes the detected disturbances (i.e., start date, end date and intensity). Individual phenotypes are first corrected for fixed effects and for the general trend of the population over time (i.e., the age for growth phenotypes, the number of days in lactation for milk yield, etc.) and smoothed using a Nadaraya-Watson's kernel regression (Nadaraya, 1964; Watson, 1964). Then, considering that animals have an elastic or plastic response (Sauvant and Martin, 2010) to disturbance that always goes in the same direction (slowing down or accelerating evolutionary dynamics over time), the Up-step identifies elements subject to at least one disturbance at the different scales by fitting, using an EM algorithm, i.e., a mixture of two Gaussian distributions to the minimum slopes of the smoothed trajectories. For levels higher than the individual level, the trajectory was obtained by taking the median value of the sub-level organization at each time point of observation (e.g., the trajectory for a given pen is obtained by taking the median of the observations of the animals in the pen at each different time). The Down-step is carried out in two stages. Firstly, it identifies all the disturbances for each element identified as having undergone at least one disturbance in the Up-step. All dates for which the local minimums of the slope are below the threshold value of the mixture model used in the Up-step are considered as disturbance start dates. The hierarchical level at which the disturbance occurs is validated if a high proportion of the sub-elements making up this hierarchical level have a close disturbance start date (obtained by clustering). The default value of this proportion is 50% in the UpDown R package. Once the hierarchical level of each disturbance has been validated, the program completes the characterization of each disturbance by identifying its end date (date of the first local minimum of the phenotype after the start of the disturbance) and its intensity (slope of the phenotype between the start and end of the disturbance). The outputs of the

UpDown package are thus a list of disturbances with their characteristics: hierarchical level affected, disturbance start and end dates, and intensity.

In addition, using the results of the UpDown package, it is possible to evaluate the individual response of each animal to the various disturbances it has undergone at the group level. We propose three statistics for this, derived from the study of the evolutionary dynamics of the smoothed corrected individual phenotype: (i) Smin, the minimum of its first derivative during disturbance exposure (i.e., between the start and the end of the disturbance or end of the observation period if the end of the disturbance is later). The latter should correspond to the primo-response of the animal exposed to a disturbance and, thus, its resistance. The greater Smin is, the greater the resistance will be; (ii) Sresil, the slope between the end of the disturbance and the end of the animal's reaction (date of a local maximum of the slope of the smoothing curve after the end of the disturbance). It should correspond to the ability of the animal to recover after an exposure to a disturbance (e.g., its resilience; the greater Sresil is, the greater the resilience will be); (iii) ABT, the area between the smoothed and theoretical (non-disturbed) trajectories (obtained by linear regression) between the start and the end of the reaction dates that should measure the global reaction of the animal, thus, its robustness. The greater ABT is, the lower the robustness will be. If the end of the reaction is not identified for an animal, neither ABT nor Sresil are computed. It should be noted that for disturbances identified at the individual level, the individual dynamic is used to characterize the disturbance, but this dynamic is also a function of the resistance and resilience of the animal. Thus, the measure of intensity is not valid for this level.

Materials

Simulated data

Longitudinal phenotypes of 6,000 individuals were simulated, as described in Le et al. (2022). Following are the main features of these simulations, described in detail in the above-mentioned article. The simulation consists in generating (i) a simplified population of five non-overlapping generations reared in different pens and batches; (ii) disturbances of different duration, intensity and starting date that occur at the individual, pen or batch level; (iii) the resistance and resilience (unobserved traits) of each individual; and (iv) the observed longitudinal trait y_{ij} for each individual i at time *j* over a 100-day observation period. For the population simulation, the generation of nonphenotyped founders comprised 12 sires and 150 dams that were randomly mated to give birth to 1,200 offspring (eight offspring per dam, sex ratio = 1/2). Among the progeny, 12 males and 150 females were randomly sampled to be the parents of the next generation. The same process was repeated for each generation. The final population comprised 6,162 individuals. To mimic a production system whereby animals are raised in small groups, animals of G1 to G5 were randomly distributed within each generation across batches and across pens within a batch. Three different types of disturbances were simulated: batch disturbances (all animals in the same batch are subjected to the same disturbance); pen (all animals in the same pen are subjected to the same disturbance); and individual (the disturbance acts on a single animal). The batches, pens and animals affected by these disturbances were randomly sampled, as was the intensity, starting point and duration of a given disturbance. The robustness of each animal was modelled using two characteristics: its resistance and its resilience. The resistance on a [0,1] scale, corresponds to the ability of an animal to minimize the direct impact of the disturbance on its performance. A resistance value of 1 indicates an animal insensible to disturbance while a resistance value of 0 corresponds to an absence of resistance, the

direct impact of the disturbance will be maximal. Resilience, on a [0,1] scale as well, corresponds to the ability to quickly return to the state before the disturbance. A resilience value of 0 indicates absence of resilience, i.e., the animal will remain in the state it was in at the end of the disturbance. On the other hand, a resilience value of 1 indicates string resilience, the animal will quickly return to its initial state. Resistance and resilience were considered constant over time for each animal and simulated using a logit genetic model considering a heritability of 0.5 on the underlying scale for both traits. The phenotype's trajectory over a 100-day period of observation for each individual was simulated using a degree-2 random regression model modified to simulate an elastic response of the animal when exposed to a disturbance. The deviation in trajectory from that which would have been observed in the absence of disturbance is a function of the intensity of the disturbance and the animal's resistance and resilience. A detailed description of the model is provided in Additional File 1. Phenotype trajectories simulated in this way increase over time, with the slope decreasing in the presence of a disturbance, as shown in Figure 2.

Simulations were carried out according to the four parameter sets described in Table 1. These differed in the percentage of disturbed elements and/or the distribution of the population in batches and pens, and thus aimed at evaluating the impact of the following on the performances of the UpDown method: (i) the percentage of disturbed elements (comparison set 1 vs. 2); (ii) disturbance at a lower level or not (set 1 vs. 3), and (iii) number of elements in a group level (set 1 vs. 4).

Using the output of the UpDown method, a genetic analysis of the robustness statistics (Smin, Sresil, ABT) obtained for animals identified as facing a disturbance occurring at the group level (batch or pen) was performed using animal models by restricted maximum likelihood method using the Average Information algorithm (AI-REML) from the ASReml software (Gilmour et al., 2015). After selection, the fixed effects included in the models were the estimated intensity for all traits and the estimated age when the disturbance started, and the estimated duration of the disturbance for Sresil and ABT. The correlation between the EBV of the different traits and the true BV for resistance and resilience were then calculated. The number of replicates used for the simulation were 1,000 for set 1 and 250 for sets 2 to 4.

Real data

Data from 5,872 pigs (3,146 Large White (LW), 1,460 Landrace and 1,266 Pietrain) of the INRAE-France Génétique Porc (UEPR, INRAE, Rennes pig experimental unit, France, 2018) in Le Rheu, France, from the year 2017 to 2019, were used in this study. Pigs were raised during the fattening period of around 100 days (from ~63d. to ~162d. of age) in pens equipped with an automatic concentrate feeder (ACF) associated with an animal weighing scale. For each visit to the ACF, individual feed intake, time spent in the ACF and BW were recorded. The animals were distributed into 52 batches and 440 different pens, with no mixing of breeds per pen; there was a median of nine pens per batch (range [6-9]) and 14 individuals per pen (range [9-15]). During this period, health data were also available. These data provided information about the treatments received by the studied pigs (animal ID, treatment date, animal main symptom, type of treatment (antibiotic, vaccine, etc.)). In total, 4,919 different pigs received at least one treatment. Based on the type of treatment (curative or preventive) and the proportion of animals in a pen or in a batch receiving the same treatment at the same time, the health data made it possible to identify disturbances occurring at the batch, pen or individual levels. In addition, the maximal daily outdoor temperature (T° in °C) and humidity (H in %) data of the facility location were collected from the 'Météo-France' website. The temperature and humidity index (THI) was calculated for each day using the following formula (Wegner et al., 2014): $THI = (1.8T^\circ + 32) - \frac{55H}{10000} \times (1.8T^\circ - 26)$. Days with a THI greater than or equal to 75 were

considered as disturbed days that impacted all batches of animals in the phenotyping station on these days. A total of 30 potential disturbances (25 different batches) at the batch scale (14 due to health problems and 16 associated with high THI), ten potential disturbances at the pen scale (nine pens) and 1,094 potential individual disturbances (812 animals) were identified.

Detection and characterization of disturbances using the UpDown package and evaluation of Smin, Sresil and ABT for animals exposed to a detected disturbance were carried out from this database using three different daily phenotypes: cumulative feed intake (CFI), body weight (BW) and feed rate (FR). The phenotypes were corrected beforehand by the median trajectory over age (over day for CFI) per breed for each phenotype and divided by their daily standard deviation in order to put phenotypes on the same scale. Heritability of each of the new traits describing the robustness of the animal (Smin, Sresil and ABT) were then estimated for LW pigs in a multitrait animal model using ASReml (Gilmour et al., 2015). For this analysis, we considered that regardless of the original phenotype used to obtain the robustness criteria (i.e., CFI, BW or FR), they correspond to the same traits. Only the robustness traits of animals considered to be undergoing a disturbance at group level (and not an individual disturbance) were taken into account for this analysis. Fixed effects included in the models were the batch, the estimated intensity of the disturbance, its estimated duration, and the age of the animal when the disturbance started. The random permanent environmental effect was not included in the models due to its low variance that was not significantly different from 0. In addition, we estimated the genetic relationship between each of the robustness criteria and the phenotypes from which they are derived to evaluate the link between robustness and production. To do so, production phenotypes must reflect the genetic potential for production only and not a combination of production and robustness (Le et al., 2022). Therefore, for animals identified as experiencing disturbances, only phenotypic records registered 4 days before the estimated starting date of the earliest disturbance were used for the analysis. For animals identified as not experiencing any disturbance, phenotypic records registered before the average estimated starting date of the detected disturbances were used for the analysis to limit the difference in the period studied for the two groups of animals. The repeated BW, FI and FR conserved phenotypes were then summarized per animal in average daily gain (ADG), average daily feed intake (ADFI) and average feed rate (AFR). The genetic correlations between robustness traits and the genetic potential for production, considering that summarized production phenotypes may correspond to different traits depending on whether the animal experienced a disturbance or not, were then estimated using a series of 3-trait multitrait animal models (one robustness trait, one summarized production trait for disturbed and non-disturbed animals). In addition to the additive genetic effects, effects included in the models for the summarized production phenotypes were the age of the animal at the date used to censor the data, and the batch.

Results

Simulated data

The sensitivity (the ability to correctly identify disturbed elements) and the specificity (the ability to correctly identify non-disturbed elements) of the UpDown method for each group level and set of simulations are provided in Table 2. The sensitivity of disturbance detection at the individual level was low, ranging from 0.23 to 0.32 depending on the set of parameters. It increased for higher levels, varying between 0.45 and 0.59 for pen disturbances, and between 0.77 and 0.88 for batch disturbances, the latter being associated with the largest standard errors. We did not observe any statistically significant differences in sensitivity depending on the set of parameters. However, it can

be noted that it tended to be lower for all hierarchical levels as the prevalence of disturbance increases (set 1 vs. 2). The specificity of the UpDown method was high regardless of the level of disturbance or the set of parameters ranging from 0.95 to 0.99. We evaluated the impact of the type of disturbance on the sensitivity of detection using a logistic regression model for each level of disturbance separately. Fixed effects included in the models were, in addition to the set of parameters, the intensity, the duration and the period of occurrence of the disturbance considered as cross-classified factors in three classes each: intensity - low (< 0.7), moderate ([0.7, 1.4]) and strong (≥ 1.4); duration - short (< 8 d.), medium ([8 d., 16 d.]) and long (\geq 16 d.); and period - early (before day 33), medium (between days 33 and 66) and late (after day 66). Odds ratio estimates of the models are provided in Figure 3. Results were similar regardless of the level at which the disturbance occurred. The main factor affecting the sensitivity of detection was the intensity of the disturbance: the stronger the intensity is, the higher the probability of detection will be. We observed a decrease in probability of detection when the disturbance is of short duration. Finally, disturbances occurring at the beginning of the control period had a slightly higher probability of detection, whereas it was lower for the batch scale when the disturbance occurred at the end of the control period. The quality of the characterization of the detected disturbances depending on the level and the set of parameters is provided in Table 3. The statistics were in the same range for the different sets of parameters. The estimated start date of the disturbance tended to be posterior to the true start date. The average lags between the estimated and true start date were ~ 2.6 and ~ 2.8 days for disturbances occurring at the batch and pen level, respectively, while it was equal to ~ 1.9 days for individual disturbances but associated with a large standard error (\sim 15). The duration of the perturbation tended to be underestimated for the different levels of disturbance in a similar manner (by ~ 1.6 days). The correlation between the true and estimated intensity of the disturbance increased with the hierarchical level of organization (on average, $\sim 0.41, 0.78$ and 0.83 for individual, pen and batch disturbances, respectively). The impact of intensity, duration and period of occurrence of the disturbance on the different statistics was evaluated using linear models. The effect of the intensity and duration of the disturbance on the lag between the estimated and true start date is provided in Figure 4. The longer the disturbance is, the greater the time interval between the estimated and true start dates will be. We also observed a slight increase in this lag when the disturbance is of low intensity. The lag between the estimated and true duration depending on the intensity and period of occurrence of the disturbance is provided in Figure 5. The underestimation of the disturbance duration was greater for disturbances of low intensity. The later the disturbance occurred, the lower the underestimation of its duration was. The impact of the period and the duration of the perturbation on the correlation between true and estimated intensity is represented in Figure 6 for the batch and pen levels. We observed a strong decrease in the correlation when the disturbance was of short duration and a slight decrease when the disturbance occurred later in the control period. Only the robustness criteria of animals having being identified as facing a disturbance occurring at the

Only the robustness criteria of animals having being identified as facing a disturbance occurring at the group level (batch or pen) were used for the genetic analysis. Therefore, only a small part of the animals in the population had phenotypes for these criteria. The proportion of animals with at least one robustness phenotype was on average 39, 68, 47 and 22% for sets 1 to 4, respectively. The results of the genetic analysis of Smin, Sresil and ABT are provided in Table 4. Heritability of the three traits ranged from low to moderate, depending on the set of parameters: from 0.12 to 0.17 for Smin; from 0.18 to 0.23 for Sresil; and from 0.13 to 0.16 for ABT. The correlation between the EBV of Sresil and the BV for resistance was low regardless of the set of parameters, ranging from -0.11 to -0.14. They were higher for the other two robustness traits, with values ranging from 0.29 to 0.33 for Smin and from 0.16 to 0.21 (in absolute value) for ABT. The correlations between EBV and BV for resilience

were low for all traits: lower than 0.05 (in absolute value) for Smin and ABT and varying between 0.09 and 0.11 for Sresil.

Real data

Applied to real data, the UpDown method identified different elements perturbed depending on the phenotype used. The corresponding venndiagram is provided in Figure 7. A total of 15 batches were considered as facing a disturbance (five detected by CFI analysis, four by BW and 11 by FR). Only four batches were detected by multiple phenotype analysis (one in common for the three phenotypes, one in common for FR and CFI and two in common between BW and FR). In addition, nine of the 15 batches detected using the UpDown package were also considered to be disturbed on the basis of meteorological and health data. A total of 54 pens were detected as facing a disturbance (27 CFI, 28 BW and 6 FR), four in common between CFI and BW and three in common between CFI and FR. Two of the detected pens also faced identified health events. Finally, a total of 1,767 individuals were considered as experiencing an individual disturbance, 52 of them were detected by the three phenotype analyses, 96 by CFI and BW, 114 by CFI and FR and 33 by FR and BW, resulting in 83% of the individual disturbances detected by one phenotype analysis only. A total of 21% of the detected individuals were also considered as experiencing a sanitary issue based on the treatment data. Robustness criteria of LW pigs detected as experiencing a group disturbance (batch or pen) were used for genetic analysis. Consequently, only 34% (1,171 animals) of the LW pigs have at least one measure of robustness. The heritability, phenotypic and genetic correlation of Smin, Sresil, ABT and summarized production traits are provided in Table 5. Heritability of the robustness traits were moderate: 0.20±0.04, 0.16±0.04 and 0.11±0.04 for Smin, Sresil and ABT, respectively. Sresil was phenotypically and genetically independent from the two other criteria, while Smin and ABT were favorably correlated (-0.76±0.14 for the genetic correlation). The heritability estimates for the summarized production phenotypes were large, varying between 0.37 and 0.55 depending on the phenotype. The phenotypic correlation between summarized production phenotypes and robustness criteria was low (< 0.18). The genetic correlation between the summarized phenotype recorded on all animals was high for ADG (0.86±0.18) and ADFI (0.91±0.10) but moderate for AFR (0.60±0.14). All the genetic correlation estimates between Sresil and the summarized phenotypes were low (all ≤ 0.14 in absolute value). The genetic correlation estimates between Smin and ADFI and AFR from nondisturbed animals were low (≤ 0.08) and slightly positive with ADG (0.25), but associated with a large standard error (0.22). When summarized production phenotypes were derived from the trajectory of animals experiencing a disturbance, the genetic correlations between Smin and production tended to be negative and significantly different from 0 for ADFI and AFR. It should be noted that genetic correlation estimates were generally associated with large standard errors. The ABT criterion was genetically independent of ADFI and AFR recorded on disturbed animals, whereas it was negatively correlated with ADF and positively with AFR recorded on animals not experiencing a disturbance. Finally, we observed a trend of favorable genetic correlation between ABT and ADG (-0.47 and -0.39 for phenotypes of non-disturbed and disturbed animals, respectively), but standard errors of the correlation estimates were large (> 0.20). Discussion

An extensive literature on the quantification of animal robustness by modeling its environmental sensitivity has emerged in recent years, thanks, in particular, to the development of high-throughput phenotyping tools that offer the possibility to characterize the dynamics of phenotype change in the face of variations in the environment. Under the assumption that animals are permanently subject to micro-variations in their environment, the fluctuation pattern of the phenotype is expected to be informative about robustness (Scheffer et al., 2018; Berghof et al., 2019). These fluctuations are

usually summarized into three different statistics derived from the difference between the observed phenotype and its modeling (i.e., residuals): the variance, autocorrelation and skewness of the residuals (Poppe et al., 2020; Homma et al., 2021; Bedere et al., 2022; Wang et al., 2022). The main difficulty in applying this method is the modeling of the theoretical trajectory of the phenotype, which must not be affected by the disturbances. This poses no problem in the case of micro-variations in the environment but is less obvious in the presence of long-term disturbances (Codrea et al., 2011; Poppe et al., 2020). This is why some authors have sought to identify and characterize disturbances before assessing the animal's response to these environmental changes. In the context of precision livestock farming, different methods have been used in livestock to identify animals whose longitudinal phenotype deviates from their expectation (Cornou et al., 2008; Maselyne et al., 2018). However, as the evolution of an individual trajectory depends on the disturbance and the robustness of the animal, such animal-by-animal approaches do not allow us to identify the disturbances suffered by the most robust animals, which will not react to the disturbances. Working at the scale of a group of animals avoids this problem, provided that the distribution of animals within a group is independent of their robustness. The UpDown method used in this study was developed around the idea (working at the group scale to detect macro variations), and although individual disturbances are identified and characterized by the method, they are not used to measure animal robustness for the reason mentioned above. It is very important to bear in mind that the UpDown method is designed to identify and measure deviations in trajectories corresponding to a response to a disturbance over the relatively long term (macro variation) and does not pretend to capture micro variation, which is correctly assessed by the residual analyses. The idea being that these long-term deviations are generally erased by the modeling of the animal's individual curve (in random regression model for instance) and therefore do not impact the residuals. The criterion currently used in the method to distinguish disturbed from nondisturbed elements is the minimum slope of the trajectory. The underlying hypothesis behind this choice is that disturbance always goes in the same direction (slowing down or accelerating evolutionary dynamics over time). Other criteria were also tested during the development of the UpDown method, derived from the analysis of the residuals (variance, skewness, autocorrelation) or from the study of the group variance over time (Garcia-Baccino et al., 2021), but resulted in poorer performance in terms of the sensitivity and specificity of disturbance detection (Le, 2022). The increase in sensitivity of the UpDown method to detect disturbances from the individual to the batch scale confirms that summarizing trajectories at the group scale level helps to detect disturbances by removing micro-variations (whether noise or not) in the studied trajectories which constitute the group. We can assume that sensitivity would have been further increased at a level above that of the batch, e.g., the breeding region. However, it must be borne in mind that the number of elements at each level must be sufficiently large to allow the mixture model to converge. The characterization of group disturbances enabled by the method can also be used as an a posteriori evaluation tool on the farm, helping to identify stress factors and levers for improvement.

Even if not statistically significant, we observed that the sensitivity of detection of the UpDown method tended to decrease with the prevalence of the disturbances (comparison set 1 vs. 2). This surprising result, given that no effect of the prevalence had been anticipated, is probably the consequence of the precorrection of the data for the general trend of the population over age. The precorrection aims at removing any natural trajectory pattern over time. In this study, this involved subtracting the median value of the phenotype for the corresponding age in the population from the observed phenotype. In set 2, several by-age median values may be affected by the disturbances since a proportion of animals exceeding 50% may be exposed to a disturbance at the same age. This leads to a higher variance of the distribution of the minimal slopes of disturbed elements in the mixture model and, thus, a greater difficulty to separate the disturbed from the non-disturbed elements. However, a

situation where 40% of the batches, the pens and the individuals are exposed to a disturbance of nonnegligible intensity is an extreme case that may not often be observed in practice, especially in breeding farms. The lack of difference in sensitivity and specificity of the UpDown method for detecting pen perturbations between sets 1 and 3 (sub-elements disturbed or not) shows the effectiveness of the method in eliminating perturbations of underlying scale elements by using the median phenotype value for a given scale and by validating the detected element in the Down-step of the method. It is important to note, however, that this would not have been the case if, by misfortune, more than 50% of the underlying elements were exposed to different perturbations, but at the same time (however, this "for validation proportion" can be modified in the UpDown package parameters). We did not highlight a significant difference in pen disturbance detection depending on the number of individuals in a pen (15 or 30 individuals, set 1 vs. 4), probably because the number of individuals remains sufficiently large in the two situations to avoid confusion between individual and pen disturbances. The extreme case of only two individuals in a pen would not have made it possible to separate the two types of disturbances (pen and individual). This is why, when we applied the UpDown method to the real pig data, we kept animals of three breeds to detect disturbances in order to avoid batches with a very small number of pens with records. As expected, the main characteristic of the disturbance influencing the sensitivity of detection was its intensity, where a disturbance of low intensity induced changes in the phenotype that were too small to be detected. A disturbance of short duration will also be less likely to be detected since the UpDown method smooths the data, eliminating deviations of excessively short duration. It should be noted that, if necessary, the smoothing parameter can be modified in the package to adapt the method to the nature of the data being analyzed. We observed that a late disturbance is less likely to be detected, especially at the batch scale. The reason is probably due to the fact that not all animals have phenotype records at the end of the test period because recording stops at 100 days of control (to mimic departure to the slaughterhouse). Consequently, for the last days of observation, the number of animals with records per pen and thus per batch were too small to permit disturbance detection by the UpDown method.

Once a disturbance is detected, the UpDown method extracts its characteristics: end and start dates, and intensity. The estimated disturbance start date is almost always slightly later than the actual date (for more than 94% of the cases for pen and batch disturbances and more than 83% of the cases for individual disturbances). The lag between the true and estimated start date was generally lower at the individual scale. This result does not mean that the start date of individual disturbances is better estimated than the start date of group level disturbances. This reduction in average lag between true and estimated start date is the consequence of trajectories identified as modified over a given period, which, in fact, do not correspond to the periods when the disturbances actually occurred, leading to a large gap between the true and estimated start dates. Indeed, for around 10% of the disturbances occurring at the individual level, the difference between the estimated and true start date (absolute value) was greater than 10 days, while this percentage was less than 1% for the disturbances occurring at the pen or batch scale. This is also reflected in the large standard deviation associated with the average gap between the start and end of the disturbance identified at the individual scale, while the median value of this gap was similar for all disturbance scales. The tendency to underestimate the duration of the disturbance is directly linked to different estimation biases for disturbance start and end dates. Both are too late estimated but the end date of the disturbance is generally more accurately estimated than the start date (< 1 day/true end date). The later estimate of the disturbance start date (and thus underestimation of the duration) was probably the consequence of trajectory smoothing (the minimal value of its first derivative being used to localize the start of the disturbance), which is all the more important the longer the disturbance duration and the lower the disturbance intensity are. When the disturbance occurs late, the duration of the disturbance is less underestimated since these

disturbances are generally shorter (9.9 d.) than those occurring earlier (12.7 d.) since the data is censored at 100 days (duration of the test period). The main factor affecting the estimation of the intensity of the disturbance is its duration. We noted that in the case of short duration, the intensity of disturbances of high true intensity tended to be underestimated. Because the smoothed trajectory cannot fall abruptly and rise rapidly over a short period, the smoothed trajectory will have a smoother evolution than the underlying noiseless trajectory in the event of a short disturbance and, therefore, a lower downward slope. To sum up, the main performance shortcomings of perturbation characterization using the UpDown method are linked to data smoothing. The default value of the bandwidth for the Nadaraya-Watson's smoothing used in the UpDown package, tuned to maximize sensitivity and specificity of the method based on our simulated data, is $n^{1/4}$ where n is the number of time points. This value can be modified by the user (the R shiny App 'UpDownApp' provided in the package UpDown can help in the choice of the best smoothing parameter). However, a bandwidth value that is too small will lead to high oscillations of the smoothing curve and, thus, false detection of disturbed elements (i.e., a reduction in the method's specificity), whereas, on the other hand, a value that is too high will lead to an over-smoothing and the algorithm might not detect the disturbances (i.e., reduction of the sensitivity).

The three criteria: Smin, Sresil and ABT, extracted from the analysis of the trajectories of animals undergoing a detected disturbance, should reflect the resistance, resilience and a combination of both (the robustness). Their estimated heritability were low to moderate, far from the 0.50 values used to simulate the underlying resistance and resilience potentials because these heritabilities are not comparable. Indeed, resistance and resilience are concepts that are not directly measurable. They are only expressed when the animal is exposed to a disturbance through a change in the trajectory of an observed phenotype. The variability of this change is linked to the environmental variability of the production phenotype itself and to the variance of the smoothing curve (Tsybakov, 2009). In other words, the residual variance of the production phenotype also contributes to the one of the measured robustness criteria. To illustrate our point, we carried out 50 additional simulations of set 3 with very low residual variance for the production phenotype (results not shown) leading to noise in the trajectory of the same order of magnitude as that observed for CFI and BW in the real data. Heritability estimates were then higher for Smin and Sresil: 0.23 and 0.43 on average, respectively; but similar for ABT (0.17). In addition, the proposed criteria are only proxies of these changes in trajectory. Furthermore, only animals detected as experiencing a group level disturbance have measured phenotypes. Nonetheless, given the non-negligible correlation between their EBV and BV for resistance, the two criteria, Smin and ABT, can provide an indication about the genetic potential of the animal in terms of resistance. Based on the simulation results, Sresil was the phenotype that gave the higher accuracy for resilience and thus remains the best proxy for this trait. Nonetheless, the accuracy was low. This may be because it is necessary to estimate the animal's end-of-response date to determine this slope, which is not feasible when the disturbance occurs late. Furthermore, this estimated slope depends on resilience as well as on the difference that exists at each time point between the observed phenotype and the value it should have had in the absence of disturbance, which may tend to add noise to the measurement.

When applying the UpDown method to real data, we standardized the longitudinal phenotypes (BW, CFI, FR) in order to obtain disturbance characteristics and Smin, Sresil and ABT on the same scale, regardless of the longitudinal phenotype used. The aim of this is to make it possible to analyze the criteria as the same trait and thus increase the number of animals with Smin, Sresil or ABT records to avoid bias in the heritability estimates. Nonetheless, this pretreatment of the data may have slightly reduced the sensitivity of the UpDown method. The UpDown method detected different disturbances

depending on the phenotype used. This is not surprising since the resource acquisition and allocation strategies (Friggens et al., 2017) that animals implement in response to disturbance may depend on the type of disturbance and the animal (Ben Abdelkrim et al., 2021). For instance, in response to heat stress, animals can reduce their feed intake, feeding rate and/or their activity, which, in turn, have or do not have consequences on their ADG (Gourdine et al., 2019; Mayorga et al., 2019). Nevertheless, we would have hoped for more concordance between analyses of different phenotypes. This would perhaps have been the case if the data analyzed had been from commercial farms where the environment is less well controlled (Lenoir et al., 2022) and where disturbances of high intensity, with consequences on multiple phenotypes, can be expected. In the experimental farm considered in this study where the environment is highly controlled, so that the disturbances encountered were a priori of low intensity, the impact of disturbances on multiple phenotypes is less obvious. In the same vein, the concordance with the disturbances suggested by the analysis of treatment and climate data was low as well. When a treatment takes place in such facilities, it is usually early before the spread of the disease and the animals have therefore not had the time to be highly exposed (more preventive than curative treatment). This seems to be confirmed by the results obtained. Indeed, 54% of the batches that received a curative treatment were identified as disturbed by the UpDown method, but at least one pen of each of the five undetected batches receiving a treatment has been identified as disturbed (1, 4, 2, 1 and 2 pens of these five batches, respectively). In addition, at least one animal of each of the seven pens considered as experiencing a sanitary challenge based on treatment data but not detected as being a disturbed pen using the UpDown method, was detected by the UpDown method as facing an individual disturbance. The temperature and humidity data used in the study were measured outside the facility and may not accurately reflect the environmental conditions inside the facility. That may explain why only three (21%) of the batches considered as having suffered thermal stress based on climate data were also detected by the UpDown method.

Genetic parameters were estimated considering that robustness criteria correspond to the same trait regardless of the longitudinal phenotype used to obtain it. We tried to investigate this hypothesis by performing additional analysis. We estimated the genetic correlation for each robustness criterion, considering the trait as different according to the type of longitudinal phenotype used to obtain it (CFI, BW, FR), through a series of two-trait models. It was difficult to get all the models to converge, and estimates of genetic correlations were associated with large standard errors, thus results should be interpreted with caution. The genetic correlation between robustness traits obtained with BW or CFI were >0.9 for the 3 robustness criteria, and varied between 0.40 and 0.71 between those obtained with FR and those obtained with the other phenotypes, depending on the criterion considered. It would therefore seem that the robustness traits obtained with BW and CFI correspond to the same trait, although this is less obvious for those obtained with FR. Heritabilities obtained for the three criteria were low to moderate, and tended to fall within the highest heritability values reported in the literature for different robustness/resilience proxies. The heritabilities based on the study of the residuals were reported to be generally low (< 0.15 (Berghof et al., 2019)), but recent studies have reported moderate values of heritability for the logarithm of the residual variance ($\simeq 0.20$ (Poppe et al., 2020)) and RMSE ($\simeq 0.25$ (Putz et al., 2019)). Heritability of the area-between-curves criterion proposed by Revilla et al. (2022), which corresponds more or less to our ABT criterion, was low (0.03-0.04), as were the robustness scores proposed by Lenoir et al. (2022) based on routinely recorded phenotypes at the end of the growing period in pig ($\simeq 0.09$). Provided that Smin was, as anticipated, a proxy of the resistance and Sresil a proxy of the resilience, we found no genetic correlation between these two components of the robustness. To our knowledge, there is no study in the literature that has estimated this genetic correlation. The heritabilities obtained on the real data were greater than those obtained

In order to evaluate the relationship between robustness and production, we used summarized production phenotypes derived from undisturbed longitudinal phenotypes, i.e., recorded before any disturbance was detected. The summarized phenotypes must correspond to the same criterion regardless of whether the animals are exposed to a disturbance or not and must reflect the true production potential of the animals. This seems to be the case for ADG and ADFI for which the genetic correlations between the two situations (disturbed or not) were high, but not for AFR for which the correlation was only 0.60. A slightly weaker correlation for a trait may be due to the fact that: (i) the sensitivity of UpDown based on this trait is lower than expected; (ii) the start date of the disturbance estimated by UpDown on this trait is much later than the true start date. Consequently, the data used to calculate summarized phenotype comprised disturbed observations; and/or (iii) the trait considerably changes with age. It is not possible to evaluate the true sensitivity of UpDown based on FR. Nevertheless, we suspect that the sensitivity of pen disturbance detection using FR was low, since the number of pens detected by FR is almost 5 times lower than that detected by CFI or BW. This may be linked to the quality of the measurement, which is probably less accurate than that of ADG or FI because it accumulates the uncertainties of the different variables used to calculate it: FI and time spent at the ASF per visit (in addition an animal can remain in the ASF without eating resulting in an error in the calculation of FR). For disturbances detected using FR and another production phenotype, the start dates of disturbance estimated using FR were generally earlier (3.4 days on average) than those estimated using the other phenotypes, which does not support the second hypothesis but is consistent with the intuition that FR reacts early to disturbance. Multiple trait animal models applied to the by-period (4 distinct periods of 25days each: [1,25], [26,50], [51,75], [76,100]) summarized phenotypes of animals not detected as exposed to any disturbance did not provide any arguments in favor of the third hypothesis, the average genetic correlation between successive periods being in the same range for ADG (0.85), AFI (0.91) than AFR (0.94). The phenotypic correlation between the summarized production phenotypes were in line with those reported by Rauw et al. (2006), strong correlation between ADG and FI, and positive correlation between FR and the other production traits. It is difficult to draw conclusions from genetic correlation estimates between robustness criteria and production because of their large standard errors. It is nevertheless interesting to note the trend towards very low genetic correlations between Smin and production traits, which would indicate, provided that Smin is a good indicator of the animal's resilience, the independence between resilience and production. On the other hand, the positive correlation between Smin and ADG in the nondisturbed situation (0.25), and the negative correlation between ABT and ADG indicate a favorable genetic correlation between resistance, robustness and growth, which is not consistent with the allocation theory (Friggens et al., 2017). In any case, further investigations on larger or different datasets are needed to confirm or refute these trends.

Given that the heritability of ABT was the lowest, integrating the Smin and Sresil traits into the selection objective seems to be the most appropriate approach for improving robustness. In any case, before considering it, it is still necessary to validate them as good proxies for resistance and resilience, especially for Sresil for which the results of the simulation were not favorable. A way to validate them could be to evaluate the genetic correlation between these traits and functional traits such as longevity, but that are difficult to identify in the pig production system where culling is rapid. However, longevity corresponds to robustness over a long term period of time, whereas the resistance and resilience that we have considered correspond to short term responses, and it is unclear how long and short term robustness are related (Friggens et al., 2022). The best option would be to perform two

divergent selections, one on Sresil and the second on Smin, and to evaluate the dynamic response of the descendent in the face of a well-controlled challenge.

Conclusion

This study presented a validation and an application of the UpDown method to detect and characterize disturbances and to estimate robustness of animals via three proxies. The results presented throughout this study proved the relevance of the UpDown method to reach these goals. To further evaluate the interest of this method for characterizing disturbances, it would be interesting to apply it to other real datasets under less controlled conditions and/or known/provoked disturbances and/or with other population structures. The three proposed robustness criteria are promising, but still need to be validated.

Competing interests

The authors declare that they have no competing interests

Acknowledgments

This study was funded 50% by the genetic animal division of INRAE and 50% by Alliance R&D (association composed of Axiom, Choice Genetics and Nucleus breeding companies, and of IFIP, the French institute for the pig and pork industry).

Real data were made available by the INRAE-France Génétique Porc station and IFIP.

The authors would like to thank the two anonymous reviewers for their helpful and constructive comments that greatly contributed to improving the final version of the paper.

Recei

References

Alonso, M. E., J. R. González-Montaña, and J. M. Lomillos. 2020. Consumers' Concerns and Perceptions of Farm Animal Welfare. Animals. 10:385. doi:10.3390/ani10030385.

Bedere, N., T. V. L. Berghof, K. Peeters, M.-H. Pinard-van der Laan, J. Visscher, I. David, and H. A. Mulder. 2022. Using egg production longitudinal recording to study the genetic background of resilience in purebred and crossbred laying hens. Genet. Sel. Evol. 54:26. doi:10.1186/s12711-022-00716-8.

Ben Abdelkrim, A., T. Tribout, O. Martin, D. Boichard, V. Ducrocq, and N. C. Friggens. 2021. Exploring simultaneous perturbation profiles in milk yield and body weight reveals a diversity of animal responses and new opportunities to identify resilience proxies. J. Dairy Sci. 104:459–470. doi:10.3168/jds.2020-18537.

Berghof, T. V. L., M. Poppe, and H. A. Mulder. 2019. Opportunities to Improve Resilience in Animal Breeding Programs. Front. Genet. doi:10.3389/fgene.2018.00692. Available from: https://www.frontiersin.org/articles/10.3389/fgene.2018.00692/full

Calus, M. P., P. Bijma, and R. F. Veerkamp. 2004. Effects of data structure on the estimation of covariance functions to describe genotype by environment interactions in a reaction norm model. Genet Sel Evol. 36:489–507. doi:10.1051/gse:2004013.

Codrea, M. C., S. Højsgaard, and N. C. Friggens. 2011. Differential smoothing of time-series measurements to identify disturbances in performance and quantify animal response characteristics: An example using milk yield profiles in dairy cows. J. Anim. Sci. 89:3089–3098. doi:10.2527/jas.2010-3753.

Cornou, C., J. Vinther, and A. R. Kristensen. 2008. Automatic detection of oestrus and health disorders using data from electronic sow feeders. Livest. Sci. 118:262–271. doi:10.1016/j.livsci.2008.02.004.

David, I., Le, Vincent, and Rohmer, Tom. 2023. UpDown -An R Package to Identify and Characterize Disturbances from Longitudinal Observations. Available from: https://hal.inrae.fr/hal-04246616

Falconer, D. S. 1990. Selection in different environments: effects on environmental sensitivity (reaction norm) and on mean performance. Genet. Res. 56:57–70. doi:doi:10.1017/S0016672300028883.

Friggens, N. C., I. Adriaens, R. Boré, G. Cozzi, J. Jurquet, C. Kamphuis, F. Leiber, I. Lora, T. Sakowski, J. Statham, and Y. De Haas. 2022. Resilience: reference measures based on longer-term consequences are needed to unlock the potential of precision livestock farming technologies for quantifying this trait. Peer Community J. 2:e38. doi:10.24072/pcjournal.136.

Friggens, N. C., F. Blanc, D. P. Berry, and L. Puillet. 2017. Review: Deciphering animal robustness. A synthesis to facilitate its use in livestock breeding and management. Animal. 11:2237–2251. doi:10.1017/S175173111700088X.

Garcia-Baccino, C. A., C. Marie-Etancelin, F. Tortereau, D. Marcon, J.-L. Weisbecker, and A. Legarra. 2021. Detection of unrecorded environmental challenges in high-frequency

recorded traits, and genetic determinism of resilience to challenge, with an application on feed intake in lambs. Genet. Sel. Evol. 53:4. doi:10.1186/s12711-020-00595-x.

Gilmour, A., B. Gogel, B. Cullis, S. Welham, and R. Thompson. 2015. ASReml user guide Release 4.1 Structural Specification. VSN Int. Ltd Hemel Hempstead HP1 1ES UK Wwwvsnicouk. Available from: https://www.hpc.iastate.edu/sites/default/files/uploads/ASREML/UserGuideStructural.pdf

Gourdine, J.-L., J. Riquet, R. Rosé, N. Poullet, M. Giorgi, Y. Billon, D. Renaudeau, and H. Gilbert. 2019. Genotype by environment interactions for performance and thermoregulation responses in growing pigs1,2. J. Anim. Sci. 97:3699–3713. doi:10.1093/jas/skz245.

Homma, C., K. Hirose, T. Ito, M. Kamikawa, S. Toma, S. Nikaido, M. Satoh, and Y. Uemoto. 2021. Estimation of genetic parameter for feed efficiency and resilience traits in three pig breeds. Animal. 15:100384. doi:10.1016/j.animal.2021.100384.

Knap, P. W. 2005. Breeding robust pigs. Aust. J. Exp. Agric. 45:763–773. doi:DOI:10.1071/EA05041.

Knap, P. W., and A. Doeschl-Wilson. 2020. Why breed disease-resilient livestock, and how? Genet. Sel. Evol. 52:60. doi:10.1186/s12711-020-00580-4.

Kolmodin, R., E. Strandberg, P. Madsen, J. Jensen, and H. Jorjani. 2002. Genotype by environment interaction in Nordic dairy cattle studied using reaction norms. Acta Agric Scand Sect Anim Sci. 52:11–24.

Le, V. G.-N. 2022. Nouvelle mesure de la robustesse des animaux d'élevage par utilisation des données de phénotypage haut-débit. Available from: http://www.theses.fr/2022INPT0087/document

Le, V., T. Rohmer, and I. David. 2022. Impact of environmental disturbances on estimated genetic parameters and breeding values for growth traits in pigs. Animal. 16:100496. doi:10.1016/j.animal.2022.100496.

Lenoir, G., L. Flatres-Grall, N. C. Friggens, and I. David. 2022. Robustness scores in fattening pigs based on routinely collected phenotypes: determination and genetic parameters. J. Anim. Sci. 100:skac157. doi:10.1093/jas/skac157.

Maselyne, J., A. Van Nuffel, P. Briene, J. Vangeyte, B. De Ketelaere, S. Millet, J. Van Den Hof, D. Maes, and W. Saeys. 2018. Online warning systems for individual fattening pigs based on their feeding pattern. Biosyst. Eng. 173:143–156. doi:10.1016/j.biosystemseng.2017.08.006.

Mayorga, E. J., D. Renaudeau, B. C. Ramirez, J. W. Ross, and L. H. Baumgard. 2019. Heat stress adaptations in pigs. Anim. Front. 9:54–61. doi:10.1093/af/vfy035.

Nadaraya, E. A. 1964. On Estimating Regression. Theory Probab. Its Appl. 9:141–142. doi:10.1137/1109020.

Nguyen-Ba, H., J. van Milgen, and M. Taghipoor. 2020. A procedure to quantify the feed intake response of growing pigs to perturbations. animal. 14:253–260. doi:10.1017/S1751731119001976.

Pasqui, M., and E. Di Giuseppe. 2019. Climate change, future warming, and adaptation in Europe. Anim. Front. 9:6–11. doi:10.1093/af/vfy036.

Phocas, F., J. Bobe, L. Bodin, B. Charley, J. Y. Dourmad, N. C. Friggens, J. F. Hocquette, P. Y. Le Bail, E. Le Bihan-Duval, P. Mormède, P. Quéré, and F. Schelcher. 2014. Des animaux plus robustes : un enjeu majeur pour le développement durable des productions animales nécessitant l'essor du phénotypage fin et à haut débit. INRAE Prod. Anim. 27:181–194. doi:10.20870/productions-animales.2014.27.3.3066.

Poppe, M., R. F. Veerkamp, M. L. van Pelt, and H. A. Mulder. 2020. Exploration of variance, autocorrelation, and skewness of deviations from lactation curves as resilience indicators for breeding. J. Dairy Sci. 103:1667–1684. doi:10.3168/jds.2019-17290.

Putz, A. M., J. C. S. Harding, M. K. Dyck, F. Fortin, G. S. Plastow, J. C. M. Dekkers, and PigGen Canada. 2019. Novel Resilience Phenotypes Using Feed Intake Data From a Natural Disease Challenge Model in Wean-to-Finish Pigs. Front. Genet. 9:660. doi:10.3389/fgene.2018.00660.

R Core Team. 2021. R: A language and environment for statistical computing. Available from: https://www.R-project.org/

Rauw, W. M., and L. Gomez-Raya. 2015. Genotype by environment interaction and breeding for robustness in livestock. Front. Genet. 6. doi:10.3389/fgene.2015.00310. Available from: http://journal.frontiersin.org/Article/10.3389/fgene.2015.00310/abstract

Rauw, W. M., J. Soler, J. Tibau, J. Reixach, and L. Gomez Raya. 2006. Feeding time and feeding rate and its relationship with feed intake, feed efficiency, growth rate, and rate of fat deposition in growing Duroc barrows1. J. Anim. Sci. 84:3404–3409. doi:10.2527/jas.2006-209.

Revilla, M., N. C. Friggens, L. P. Broudiscou, G. Lemonnier, F. Blanc, L. Ravon, M. J. Mercat, Y. Billon, C. Rogel-Gaillard, N. Le Floch, J. Estellé, and R. Muñoz-Tamayo. 2019. Towards the quantitative characterisation of piglets' robustness to weaning: a modelling approach. Animal. 13:2536–2546. doi:10.1017/S1751731119000843.

Revilla, M., L. Guillaume, F.-G. Loïc, M.-T. Rafael, and N. C. Friggens. 2022. Quantifying growth perturbations over the fattening period in swine via mathematical modelling. Peer Community J. 2:e9. doi:10.24072/pcjournal.82.

Sadoul, B., O. Martin, P. Prunet, and N. C. Friggens. 2015. On the use of a simple physical system analogy to study robustness features in animal sciences. PLoS One. 10:e0137333. doi:10.1371/journal.pone.0137333.

Sauvant, D., and O. Martin. 2010. Robustesse, rusticité, flexibilité, plasticité... les nouveaux critères de qualité des animaux et des systèmes d'élevage: définitions systémique et biologique des différents concepts. INRAE Prod. Anim. 23:5–10. doi:10.20870/productions-animales.2010.23.1.3280.

Scheffer, M., J. E. Bolhuis, D. Borsboom, T. G. Buchman, S. M. W. Gijzel, D. Goulson, J. E. Kammenga, B. Kemp, I. A. van de Leemput, S. Levin, C. M. Martin, R. J. F. Melis, E. H. van Nes, L. M. Romero, and M. G. M. Olde Rikkert. 2018. Quantifying resilience of humans and other animals. Proc. Natl. Acad. Sci. 115:11883–11890. doi:10.1073/pnas.1810630115.

Tsybakov, A. B. 2009. Introduction to nonparametric estimation. Springer, New York; London.

UEPR, INRAE, Rennes pig experimental unit, France. 2018. doi:10.15454/1.557241481185847E12.

Wang, A., L. F. Brito, H. Zhang, R. Shi, L. Zhu, D. Liu, G. Guo, and Y. Wang. 2022. Exploring milk loss and variability during environmental perturbations across lactation stages as resilience indicators in Holstein cattle. Front. Genet. 13:1031557. doi:10.3389/fgene.2022.1031557.

Watson, G. S. 1964. Smooth Regression Analysis. Sankhyā Indian J. Stat. Ser. 1961-2002. 26:359–372. doi:https://www.jstor.org/stable/25049340.

Wegner, K., C. Lambertz, G. Daş, G. Reiner, and M. Gauly. 2014. Climatic effects on sow fertility and piglet survival under influence of a moderate climate. Animal. 8:1526–1533. doi:10.1017/S1751731114001219.

Receit

List of figures

Figure 1: Theoretical elastic changes in the state of an animal facing a disturbance When the disturbance appears (start line), the importance of change in state will depend on the resistance of the animal and the intensity of the disturbance. As soon as the animal has quit its initial state, the resilience capacity is involved in order to limit the effect of the perturbing factor so that it can return to its initial state.

Figure 2: Illustration of the simulated phenotypes (points) associated with their smoothing curve for three animals

Animal 1 did not face any disturbance; animals 2 (resistance=0.5170, resilience=0.5490) and 3 (resistance=0.3650, resilience=0.5423) belonged to the same pen where a disturbance occurred between days 60 and 77 (vertical blue lines). The trajectory of animal 3 was more deviate than that of animal 2 due to its lower resistance.

Figure 3: Forest plot of the factors impacting the sensitivity of disturbance detection

Intensity: intensity of the disturbance in 3 classes, low (< 0.7), moderate (MOD) ([0.7,1.4]) and strong (\geq 1.4). Duration: duration of the disturbance in 3 classes, short (< 8 *d*.), medium (MED)([8 *d*.,16 *d*.]) and long (\geq 16 *d*.). Period: period where the disturbance occurs in 3 classes, early (before day 33), medium (MED) (between days 33 and 66) and late (after day 66)

Figure 4: Impact of disturbance duration and intensity on the estimation of its start date

Intensity: intensity of the disturbance in 3 classes, low (< 0.7), moderate (MOD) ([0.7,1.4[) and strong (\geq 1.4). Duration: duration of the disturbance in 3 classes, short (< 8 *d*.), medium (MED)([8 *d*.,16 *d*.[) and long (\geq 16 *d*.)

Figure 5: Impact of disturbance intensity and occurrence period on the estimation of its duration

Intensity: intensity of the disturbance in 3 classes, low (< 0.7), moderate (MOD) ([0.7,1.4[) and strong (\geq 1.4). Period: period where the disturbance occurs in 3 classes, early (before day 33), medium (MED) (between days 33 and 66) and late (after day 66)

Figure 6: Impact of disturbance duration and period of occurrence of the estimation of its intensity

Duration: duration of the disturbance in 3 classes, short (< 8 *d*.), medium (MED)([8 *d*.,16 *d*.]) and long (\geq 16 *d*.). Period: period where the disturbance occurs in 3 classes, early (before day 33), medium (MED) (between days 33 and 66) and late (after day 66)

Figure 7: Venndiagram of the elements detected at the different hierarchical levels depending on the phenotype used

BW: Body weight; CFI: cumulative feed intake; FR: feeding rate; th: theoretical disturbed elements based on THI and medical treatments

Accepted Manuscrip

Tables

Table 1: Sets of parameters used for the simulation

	Set1	Set2	Set3	Set4
#batches per generation	8	8	8	4
#pens within batch	10	10	10	10
#animals per pen	15	15	15	30
% batches disturbed	20	40	20	20
% pens disturbed	20	40	20	20
% animals disturbed	20	40	0	20

same population for each set: 1,200 phenotyped individuals per generation, five generations

NS

Set	batch	pen	individual			
		Sensitivity				
set1	0.82 ± 0.15	0.54 ± 0.07	0.32 ± 0.04			
set 2	0.77 ± 0.13	0.45 ± 0.07	0.23 ± 0.06			
set 3	0.83 ± 0.14	0.55 ± 0.07	-			
set 4	0.88 ± 0.17	0.59 ± 0.1	0.30 ± 0.05			
	Specificity					
set1	0.99 ± 0.02	0.98 ± 0.01	0.95 ± 0.01			
set 2	0.98 ± 0.03	0.98 ± 0.02	0.96 ± 0.01			
set 3	0.99 ± 0.02	0.99 ± 0.01	0.95 ± 0.01			
set 4	0.99 ± 0.04	0.99 ± 0.02	0.95 ± 0.01			

Table 2: Proportion of well detected elements among the disturbed elements (sensitivity) and among the undisturbed element (specificity) using the UpDown method for the different group levels and sets of parameters (mean \pm sd)

Set 1: 20% of all hierarchical levels (batch, pen, individual) exposed to disturbances, Set 2: 40% of all hierarchical levels exposed to disturbances, Set 3: 20% of batch and pen level exposed to disturbance. Set 4: 20% of all hierarchical levels exposed to disturbances, 4 batches.

Accer

set	level	Estimated-true start date	Estimated-true duration	Correlation between estimated and true intensity
1	batch	2.60 ± 1.72	-1.35 ± 3.15	0.83 ± 0.008
	pen	2.83 ± 4.88	-1.71 ± 3.57	0.77 ± 0.003
	individual	1.78 ± 14.47	-1.68 ± 5.15	0.42 ± 0.002
2	batch	2.81 ± 2.79	-1.55 ± 3.15	0.81 ± 0.008
	pen	2.93 ± 6.30	-1.74 ± 3.85	0.75 ± 0.004
	individual	1.94 ± 15.29	-1.63 ± 5.49	0.37 ± 0.002
3	batch	2.54 ± 2.46	-1.31 ± 3.03	0.84 ± 0.009
	pen	2.72 ± 4.82	-1.69 ± 3.48	0.78 ± 0.004
4	batch	2.59 ± 1.98	-1.38 ± 3.04	0.83 ± 0.014
	pen	2.78 ± 3.93	-1.65 ± 3.45	0.81 ± 0.006
	individual	1.85 ± 14.56	-1.71 ± 5.14	0.43 ± 0.002

Table 3: Performances of the UpDown method to characterize detected disturbances for the different group levels and sets of parameters (mean±sd)

Set 1: 20% of all hierarchical levels (batch, pen, individual) exposed to disturbances, Set 2: 40% of all hierarchical levels exposed to disturbances, Set 3: 20% of batch and pen level exposed to disturbance. Set 4: 20% of all hierarchical levels exposed to disturbances, 4 batches.

Accel

Set	h^2	$\rho(EBV, u_{resis})$	$\rho(EBV, u_{resil})$
		Smin	
set1	0.12 ± 0.14	0.29 ± 0.09	0.05 ± 0.10
set 2	0.12 ± 0.09	0.33 ± 0.09	0.04 ± 0.08
set 3	0.17 ± 0.17	0.30 ± 0.09	0.04 ± 0.09
set 4	0.16 ± 0.17	0.27 ± 0.12	0.03 ± 0.09
		Sresil	
set1	0.19 ± 0.24	$\textbf{-0.11} \pm 0.09$	0.09 ± 0.08
set 2	0.19 ± 0.16	$\textbf{-0.14} \pm 0.08$	0.11 ± 0.08
set 3	0.23 ± 0.26	$\textbf{-0.12} \pm 0.09$	0.10 ± 0.09
set 4	0.18 ± 0.19	-0.11 ± 0.09	0.09 ± 0.09
		ABT	
set1	0.13 ± 0.17	$\textbf{-0.17} \pm 0.09$	-0.03 ± 0.09
set 2	0.13 ± 0.14	$\textbf{-0.21} \pm 0.09$	-0.04 ± 0.08
set 3	0.16 ± 0.20	$\textbf{-0.18} \pm 0.09$	-0.03 ± 0.09
set 4	0.13 ± 0.17	-0.16 ± 0.11	-0.03 ± 0.09

Table 4: Heritability of the different robustness criteria and accuracy of their EBV for resistance and resilience (mean $\rho \pm sd$)

Smin: slope minimum, Sresil: Slope for resilience, ABT: area between trajectories.

Set 1: 20% of all hierarchical levels (batch, pen, individual) exposed to disturbances, Set 2: 40% of all hierarchical levels exposed to disturbances, Set 3: 20% of batch and pen level exposed to disturbance. Set 4: 20% of all hierarchical levels exposed to disturbances, 4 batches.

Downloaded from https://academic.oup.com/jas/advance-article/doi/10.1093/jas/skae059/7619452 by INRA - DOCUMENTATION user on 06 March 2024

Table 5: Heritability (on the diagonal), phenotypic correlation (above the diagonal) and genetic correlation (below the diagonal) of the different robustness criteria measured on LW pigs

	Smin	Sresil	ABT	ADG _{nd}	ADG _d	ADFI _{nd}	ADFI _d	AFR _{nd}	AFR _d
Smin	0.20±0.04	-0.10±0.03	-0.52±0.03		-0.12±0.03		0.18±0.03		0.09±0.03
Sresil	-0.01±0.17	0.16±0.04	0.07±0.03		-0.09 ± 0.04		0.07±0.04		0.10±0.04
ABT	-0.76±0.14	0.19±0.21	0.11±0.04		0.03±0.04		-0.03±0.04		-0.02 ± 0.04
ADG _{nd}	0.25±0.22	0.04±0.26	-0.47±0.30	0.38 ± 0.07	-	0.80±0.02		0.42±0.02	
ADG _d	-0.17±0.13	0.00±0.16	-0.39±0.22	0.86 ± 0.18	0.39±0.12		0.45±0.03		0.36±0.03
ADFI _{nd}	0.08±0.19	0.14±0.21	-0.30±0.24			0.55±0.07	-	0.57±0.02	
ADFI _d	-0.33±0.10	0.01±0.12	0.01 ± 0.17			0.91±0.10	0.37±0.09		0.81±0.02
AFR _{nd}	0.02 ± 0.18	-0.14±0.23	0.17±0.28					0.50±0.07	-
AFR _d	-0.58±0.08	-0.02±0.13	0.07±0.17					0.60±0.14	0.50 ± 0.09

Smin: slope minimum, Sresil: Slope for resilience, ABT: area between trajectories, ADG: average daily gain, ADFI: average daily feed intake, AFR: average feeding rate. Subscript nd: non disturbed, d: disturbed.

























